Author’s response to reviews

Title: Correct Recognition and Continuum Belief of Mental Disorders in a Nursing Student Population

Authors:

Lee Seng Esmond Seow (esmond_LS_SEOW@imh.com.sg)
Boon Yiang Chua (boon_yiang_chua@imh.com.sg)
Huiting Xie (hui_ting_xie@imh.com.sg)
Jia Wang (jia_wang@imh.com.sg)
Hui Lin Ong (Hui_Lin_ONG@imh.com.sg)
Edimansyah Abdin (edimansyah_abdin@imh.com.sg)
Siow Ann Chong (siow_ann_chong@imh.com.sg)
Mythily Subramaniam (mythily@imh.com.sg)

Version: 2 Date: 20 Jul 2017

Author’s response to reviews:

Reviewer reports:

Rishikesh V Behere (Reviewer 2): The authors have addressed the concerns raised adequately in the reply to reviewer comments.

However I would appreciate if the authors could add a note regarding the issue of controlling for inter observer variations and biases and power of the sample in the manuscript as well as these are important methodological issues in this study.

Reply to reviewer: Thank you for the reviewer's suggestion. We have included a note in our methods (Page 5, 6) to address the inter observer variations and power of sample in the text.